Author's response to reviews

Title: Clinical and molecular features of an infant patient affected by Leigh Disease associated to m.14459G>A mitochondrial DNA mutation: a case report.

Authors:

Dario Ronchi (darioronchi@tin.it)
Alessandra Cosi (alessandra.cosi@studenti.unimi.it)
Davide Tonduti (davidetondu@hotmail.com)
Simona Orcesi (simona.orcesi@mondino.it)
Andreina Bordoni (andreina.bordoni@unimi.it)
Francesco Fortunato (francesco.fortunato@unimi.it)
Mafalda Rizzuti (mafalda.rizzuti@studenti.unimi.it)
Monica Sciacco (monica.sciacco@policlinico.mi.it)
Martina Collotta (martina.collotta@studenti.unimi.it)
Sophie Cagdas (sophie.cagdas@aopoma.it)
Giuseppe Capovilla (pippo.capovilla@aopoma.it)
Maurizio Moggio (maurizio.moggio@unimi.it)
Angela Berardinelli (angela.berardinelli@mondino.it)
Pierangelo Veggiotti (pierangelo.veggiotti@unipv.it)
Giacomo P. Comi (giacomo.comi@unimi.it)

Version: 3 Date: 9 July 2011

Author's response to reviews:

Details of this revision (10th july 2011):

Major revisions:
- Shaded cells within the table have been removed.

Minor revisions:
- All highlighting and tracking have been removed from the manuscript.
- Visible vertical lines have been removed from the table.
- Manuscript has been carefully revised for mispelling errors taking into account suggestions provided by Reviewer 2.